Author’s response to reviews

Title: Presumed Stromal Graft Rejection After Automated Lamellar Therapeutic Keratoplasty

Authors:

Motoko Kawashima (motokok@tdc.ac.jp)
Hiroshi Mochizuki (kawatetsu@gmail.com)
Tetsuya Kawakita (kawatetsu@gmail.com)
Shin Hatoh (kawatetsu@gmail.com)
Jun Shimazaki (kawatetsu@gmail.com)
Masakazu Yamada (kawatetsu@gmail.com)

Version: 2 Date: 19 January 2007

Author’s response to reviews:

We review our manuscript from the point of reviewers, reduced references, and adding abbreviations list, an authors contribution list and an acknowledgement list.